%
% 92.02.23 ysato
%
% 93/03/19 yoshimi check
\input{guide-style}
\section[vin $@$N;H$$J}(J(vin: y$@:4F#(J)]{vin $@$N;H$$J}(J}
\label{sec:vin} \label{sec:vin-guide}

\sindex{vin}
\index{$@%K%e!<%9(J}
\index{$@%a!<%k(J}
{\tt vin}$@$O(J1987$@G/@8$^$l$NEEAm8&=c@5%K%e!<%9!&%a!<%k%j!<%@$G$"$j!"%K%e!<%9%j!<%@(J
$@$H$7$F$O!"8=:_=jFb$G:G$bB?$/$NMxMQ<T(J(93$@G/(J3$@7n8=:_$G(J100$@!A(J150$@?M(J)$@$KMxMQ$5$l$F(J
$@$$$k%D!<%k$G$9!#=i$a$FEE;R%K%e!<%9$dEE;R%a!<%k$K@\$9$k?M$+$i!"BgNL$N%K%e!<%9$d(J
$@%a!<%k$N=hM}$KGw$i$l$F$$$k?M$^$G!"$"$k$$$O!"KhF|%K%e!<%9$d%a!<%k$rFI$`?M$+$i!"(J
$@;~!9;W$$=P$7$?;~$KFI$`?M$^$G!"$=$l$>$l$K1~$8$?;H$$J}$,$5$l$F$$$k$h$&$G$9!#(J
$@$3$3$G$O!"$O$8$a$F(J{\tt vin}$@$r;H$&?M$rBP>]$K!"$=$NFCD'$H@-3J!"$=$7$F$R$HDL$j$N(J
$@;H$$J}(J($@FI$_=q$-@0M}(J)$@$r>R2p$7$^$9!#(J

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{vin$@$NFCD'(J}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
{\tt vin}$@$NBh0l$NFCD'$O!"%a!<%k$r%K%e!<%9$HF1$8$h$&$K07$($kE@$G$9!#$D$^$j!"(J
$@%a!<%k$H%K%e!<%9$r07$&$?$a$KJL!9$N%D!<%k$r3P$($kI,MW$,$"$j$^$;$s!#(J
{\tt vin}$@MxMQ<T$+$i8+$k$H!"%a!<%k(J($@%U%)%k%@(J)$@$O0l<o$N%K%e!<%9%0%k!<%W$N$h$&$K(J
$@8+$(!"%K%e!<%9$HF1$8$h$&$KFI$`$3$H$,$G$-$^$9!#%a!<%k%U%)%k%@7A<0$H$7$F$O!"(J
UNIX$@I8=`7A<0(J(MBOX$@7A<0(J)$@$*$h$S(JMH$@7A<0$NN><T$r%5%]!<%H$7$F$$$^$9$N$G!"(J
UCB mail$@$d(JMH$@$=$NB>3F<o$N%a!<%k%O%s%I%i$HLdBj$J$/6&B8$G$-$^$9!#(J
\index{MBOX$@%1%$%7%-(J@MBOX$@7A<0(J}\index{MH$@%1%$%7%-(J@MH$@7A<0(J}

$@$b$&0l$D$NFCD'$H$7$F!"%K%e!<%9%0%k!<%W$N3,AXE*$J8+J}$r%5%]!<%H$7$F$$$kE@(J
$@$,5s$2$i$l$^$9!#$3$N$?$a!"Nc$($P(J{\tt etl.*}$@A4BN$K5-;v$,Mh$F$$$kH]$+$r$R$HL\$G(J
$@CN$k$3$H$,$G$-$^$9$7!"$"$k$$$O!"Nc$($P%H%C%W%l%Y%k$G$^$:(J{\tt etl}$@$rA*$V$H!"8e$O(J
{\tt etl}$@$N2<$@$1$,A*BrBP>]$H$J$k$N$G!"A*Br$N$?$a$N%+!<%=%k0\F02s?t$,8:$i$;$^$9!#(J
$@$3$N%K%e!<%9%0%k!<%W$N3,AX$O!"MxMQ<T$,<+J,$N9%$-$J$h$&$K9=@.$9$k$3$H$,$G$-(J
$@$F!"6=L#$N$"$k%K%e!<%9%0%k!<%W$@$1$r$R$H$^$H$^$j$K$7$?$j$G$-$^$9!#(J

$@@-G=E*$K(J{\tt vin}$@$O!V=E$$!W%D!<%k$H;W$o$l$F$$$^$9$,!"(J{\tt vin}$@$HF1MM$N5!G=$rDs6!(J
$@$7$F$$$k%D!<%k$HHf$Y$l$P!"==J,9bB.$G$9!#Nc$($P!"(JMBOX$@7?%a!<%k%U%)%k%@$KBP(J
$@$7$F$O%$%s%G%C%/%9%U%!%$%k$r30IU$1$G0];}$7$F$$$k$N$G!"%U%)%k%@$,Bg$-$/$J$C(J
$@$F$b!"$=$NBg$-$5$K$[$H$s$I4X78$J$/9bB.$KN)$A>e$,$j$^$9!#$^$?!"%K%e!<%9$KBP$7(J
$@$F$O5-;v$N%X%C%@ItJ,$r(J{\tt vin}$@%f!<%64V$G6&M-$7$F%-%c%C%7%e$7$F$$$k$?$a!"(J
$@9bB.$K%K%e!<%9%0%k!<%W$KF~$k$3$H$,$G$-$^$9(J($@5-;v$N0lMw$N:n@.$,9bB.(J)$@!#(J

$@%3%^%s%IBN7O$H$7$F$O!"%+!<%=%k0\F04X78$O4pK\E*$K(J{\tt vi}$@$N%3%^%s%I$r7Q>5$7$F(J
$@$$$^$9!#$^$?!"%K%e!<%9!&%a!<%kA`:n$N%3%^%s%I$O!"(J{\tt rn}$@$d(JUCB mail$@$+$iB?$/$r(J
$@<u$17Q$$$G$$$^$9!#8D?M$4$H$N%K%e!<%9$d%a!<%k$N4IM}>pJs(J({\tt .newsrc$@$d(J.mailrc})
\index{.newsrc}\index{.mailrc}
$@$J$I$b!"(J{\tt rn}$@$d(JUCB mail$@$N$b$N$r6&M-$7$F$$$^$9!#$3$N$?$a!">o$K(J{\tt vin}$@$r;H$$B3$1$k(J
$@$N$G$O$J$/!"I,MW$H46$8$?;~$K$@$1;H$&$H$$$&;H$$J}$bL5M}$J$/$G$-$k$G$7$g$&!#(J

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{$@<B83%7%9%F%`$H$7$F$N(Jvin}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

{\tt vin}$@$OMxMQ<T%$%s%?!<%U%'%$%9$N8&5f$K$*$$$F!"%G!<%?<}=8$N$?$a$NBj:`$H$7$F(J
$@3+H/$5$l$FMh$?<B83%7%9%F%`$G$9!#$=$N$?$a!"(J{\tt vin}$@$rMxMQ$9$k$"$J$?$N%3%^%s%I(J
$@MxMQMzNr%G!<%?$O!"7QB3E*$K<}=8$5$l$^$F$$$^$9(J($@8@$&$^$G$b$J$/!"8D?M>pJs$NHk(J
$@L)$O87<i$5$l$^$9(J)$@!#MxMQ<T$N0c$$$KE,1~$7$?E,@Z$J%$%s%?!<%U%'%$%9$r<B8=$9$k!"(J
$@$H$$$&$N$,$3$N8&5f$N<gL\E*$G$"$j!"$3$N$h$&$K<}=8$5$l$?%G!<%?$O!"$d$,$F(J
$@<B:]$K$"$J$?$N(J{\tt vin}$@$r$"$J$?9%$_$K<+F0E*$KJQ?H$5$;$k$N$K;H$o$l$k$+$bCN$l(J
$@$^$;$s!#(J

$@$^$?!"(J{\tt vin}$@$O8=:_$G$b2~D{$,B3$1$i$l!"<B83E*$J5!G=$,AH$_9~$^$l!"$=$7$F;~$K(J
$@$h$j%P%0$,F~$j9~$`$3$H$b$"$j$^$9!#:n@.<TB&$G40A4$J%A%'%C%/$r$9$k$3$H$OIT2D(J
$@G=$G!"B?MM$JMxMQ<T$K$h$kB?MM$J;H$o$lJ}$r7P$F=i$a$F8!=P$5$l$k%P%0$b$"$j$^$9!#(J
{\tt vin}$@$,0[>o;vBV$K4Y$C$?;~$K$O!"(J{\tt vin}$@<+?H$,%P%0$NJs9p%a!<%k$r:n@.<T08$KAw?.$7$^$9!#(J
$@$=$N;~!"0[>o$N<+8J?GCG$N$?$a$KMxMQ<T$K$7$P$i$/(J($@#1J,DxEY(J)$@BT$C$F$b$i$&$3$H$K(J
$@$J$j$^$9!#(J

$@$3$l$i$N0UL#$G!"(J{\tt vin}$@MxMQ<T$O1"$K!">o$K<B83$K;22C$7$F$$$k(J($@$5$;$i$l$F$$$k(J)
$@$3$H$r$4N;>52<$5$$!#$=$7$F!"M[$K!"@Q6KE*$K;22C$7$F2<$5$k$3$H$r4?7^$7$^$9!#(J
{\tt vin}$@$O8=:_$^$G!"(J3400$@DL$rD6$($k%P%0%l%]!<%H$H!"B??t$N2~A1Ds0F$K;Y$($i$l$F(J
$@@.D9$7$FMh$?$N$G$9!#(J
{\tt vin}$@$N%$%s%9%H!<%k$dMxMQ$K4X$9$k>pJs$O!"(J{\tt etl.mail-lists.vin}$@$H$$$&%K%e!<%9(J
$@%0%k!<%W$G8r49$5$l$F$$$^$9!#(J{\tt vin}$@$K4X$7$F$N<ALd$dMW5a$,$"$k;~$K$O!"(J
$@$3$N%0%k!<%W$KEj9F$7$F2<$5$$!#$^$?!"$3$N%0%k!<%W$O=j30$N(J{\tt vin}$@MxMQ<T(J
$@$H$b@\B3$5$l$F$$$F!"(J{\tt Distribution: fj}$@$H$7$FEj9F$9$k$H!"=j30$K$bG[(J
$@Aw$5$l$^$9!#(J\index{etl.mail-lists.vin}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{$@4pK\E*$J;H$$J}(J}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

$@0J2<$K!"(J{\tt vin}$@$N4pK\E*$J;H$$J}$r@bL@$7$^$9!#$3$3$K=R$Y$k%3%^%s%I$GBgDq$NA`:n(J
$@$O2DG=$G$9$,!"$h$j8zN(E*$KMxMQ$9$k$?$a$K$O!"(JUNIX$@>e$N%*%s%i%$%s%^%K%e%"(J
$@%k(J({\tt man vin}$@$G8+$($k(J)$@!"$*$h$S(J{\tt vin}$@$N%*%s%i%$%s%I%-%e%a%s%H$r;2(J
$@>H$7$F2<$5$$!#(J
$@ITL@$JE@$O%K%e!<%9%0%k!<%W(J{\tt etl.mail-lists.vin}$@$G<ALd$7$F2<$5$$!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{description}
\itemsep=1mm
\item{\large \bf (1) $@5/F0(J}

{\tt vin}$@$r5/F0$9$k$K$O!"(JUNIX$@$N%3%^%s%I%l%Y%k$G!"0J2<$N$h$&$KF~NO$7$F2<$5$$!#(J

    {\tt \% vin}

$@%^%7%s$N@-G=$dIi2Y>uBV!"$"$k$$$O8D?M$N4D6-@_Dj$K$h$j$^$9$,!"DL>o$O?tIC!A(J
$@?t==IC$GN)$A>e$,$j$^$9!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (2) $@:G>.8B$N%3%^%s%I(J}

$@=i$a$F(J{\tt vin}$@$r5/F0$7$?MxMQ<T$KBP$7$F$O!"(J{\tt vin$@$O(Jcosmos.vin.manual}
$@$NCf$N0J2<$N$h$&$J5-;v$+$i3+;O$7$^$9!#(J
$@$3$3$G$O!"(J{\tt vin}$@$rMxMQ$9$k$?$a$NI,?\$+$D:G>.8B$N%3%^%s%I%;%C%H$r@b(J
$@L@$7$F$$$^$9(J
($@$3$N>pJs$O!"?o;~8e=R$N%*%s%i%$%s%X%k%W(J\verb+^Hi+$@$G8+$i$l$^$9(J)$@!#(J
$@$3$N5-;v$NCf$G(J{\tt q}$@$HBG$D$H!"DL>o$N3+;O2hLL(J($@%H%C%W%l%Y%k(J)$@$K0\$j$^$9!#(J
\dense
\begin{verbatim}
$@!!!!!!(J------------------------------------------------------------------------
$@!!!!!!!!(J8:$@!!(Jysato$@!!(J9/28-23:42 (32)$@!!$O$8$a$F#v#i#n$r;H$&?M$N$?$a$K(J
$@!!!!!!(J------------------------------------------------------------------------
$@!!!!!!!!#v#i#n$O!"%+!<%=%k$N>e2<0\F0(J($@#j!"#k(J)$@!"%0%k!<%W$d5-;v$NA*Br$HC&=P(J($@#i!"#o(J)
$@!!!!!!!!%3%^%s%I!"$*$h$S=*N;(J($@#q!"#Q(J)$@$@$1$G!"0l1~;H$&$3$H$,$G$-$^$9!#(J
$@!!!!!!!!(J
$@!!!!!!!!!!!!9TC10L$N0\F0!!#j!"#k!!2<$N9T$X(J(j)$@!">e$N9T$X(J(k)$@!!(J($@%+!<%=%k%-!<$b2D(J)
$@!!!!!!!!!!!!JGC10L$N0\F0!!(JSP$@!"#b!!<!$NJG$X(J(SPace)$@!"A0$NJG$X(J(b)
$@!!!!!!!!!!!!3,AX4V$N0\F0!!#i!"#o!!$=$l$KF~$k(J(in)$@!"$3$3$+$i=P$k(J(out)
$@!!!!!!!!!!!!#v#i#n$N=*N;!!#q!"#Q!!0lC6:G>e0L$KLa$C$F=*N;(J(q)$@!"B(:B$K=*N;(J(Q)
$@!!!!!!!!(J
$@!!!!!!!!(J  $@#v#i#n$r5/F0$9$k$H!":G=i$K%0%k!<%W$N:G>e0L$N3,AX$,%j%9%H$5$l$^$9!#(J
$@!!!!!!!!FI$_$?$$%0%k!<%W$K#j$d#k$G%+!<%=%k$r$"$o$;!"#i$rBG$F$P$=$N%0%k!<%W$K(J
$@!!!!!!!!F~$j$^$9!#(J
$@!!!!!!!!!!$3$l$r7+$jJV$7$F$$$/$H!"$d$,$F5-;v$N%j%9%H$,8=$l$^$9!#5-;v$N%j%9%H$N>e(J
$@!!!!!!!!$GFI$_$?$$5-;v$K%+!<%=%k$r$"$o$;!"#i$G$=$N5-;v$KF~$j$^$9!#(J
$@!!!!!!!!!!5-;vCf$N%a%C%;!<%8(JID$@$N>e$G#i$rBG$D$H$=$N5-;v$KF~$j$^$9!#85$N5-;v$KLa$k(J
$@!!!!!!!!$K$O(JRET$@$rBG$A$^$9!#(J
$@!!!!!!!!(J  $@5U$K5-;v$d%0%k!<%W$r>e$XH4$1=P$7$F9T$/$?$a$N%3%^%s%I$,#o$G$9!#(J
$@!!!!!!!!(J
$@!!!!!!!!!!(JSPC$@$*$h$S#b$O%Z!<%8$a$/$j$G$9!#=*N;$O#q$^$?$O#Q$G$9!##Q$G=*$k$H!"(J
$@!!!!!!!!<!$N#v#i#n$OA02s$N=*N;;~$N>l=j$+$i;O$^$j$^$9!#(J
$@!!!!!!!!(J
$@!!!!!!!!!!%3%^%s%I0lMw$O(J^Hs$@$G8+$i$l$^$9!#>\$7$$@bL@$O(Jcosmos.vin.manual$@!"%P!<(J
$@!!!!!!!!%8%g%s%"%C%W>pJs$O(Jcosmos.vin.history$@$H$$$&%0%k!<%W$K$"$j$^$9!#(J
$@!!!!!!!!:F$S$3$N%$%s%H%m%@%/%7%g%s$r8+$?$$$H$-$O!"(J^Hi$@$HBG$C$F2<$5$$!#(J
$@!!!!!!(J------------------------------------------------------------------------
\end{verbatim}
\sparse
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (3) $@%0%k!<%W3,AX$H%3%^%s%I%l%Y%k$*$h$SI=<(7A<0(J}

\ref{nandakorya1}$@$N$h$&$K!"%K%e!<%9%0%k!<%W$O3,AXE*$K9=@.$5$l$F$$$F$$$^$9!#(J
$@$3$l$KBP1~$7$F!"(J\ref{nandakorya2}$@$K<($9$h$&$K!"(J{\tt vin}$@$K$O0J2<$N#3<oN`$N%3%^%s%I%l%Y%k$,$"$j$^$9!#(J
\begin{verbatim}
$@!!!!!!!!%0%k!<%W%j%9%H%l%Y%k!'8=:_$N%0%k!<%W$N2<0L%0%k!<%W$,Ns5s$7$FI=<($5$l$k(J
$@!!!!!!!!5-;v%j%9%H%l%Y%k!'!!!!8=:_$N%0%k!<%W$N5-;v$,(J1$@5-;v(J1$@9T$GNs5s$7$FI=<($5$l$k(J
$@!!!!!!!!5-;v%l%Y%k!'!!!!!!!!!!8D!9$N5-;v$NFbMF$,I=<($5$l$k(J
\end{verbatim}
$@3F%l%Y%k$G$O07$&BP>]$,0[$J$j$^$9$+$i!"%3%^%s%I$N0UL#$,<c430[$J$j$^$9!#(J
$@$?$@$7!">e5-(J(2)$@$NI,?\%3%^%s%I%;%C%H$r4^$a!"B?$/$N%3%^%s%I$O!"$I$N%l%Y%k(J
$@$G$b6&DL$N0UL#$r;}$C$F$$$^$9!#(J

{\tt vin}$@$NI=<(2hLL$N$&$A!"0lHV>e$N9T$r%?%$%H%k9T!"0lHV2<$N9T$r%a%C%;!<%89T$H(J
$@8F$S$^$9!#%?%$%H%k9T$K$O8=:_$N%0%k!<%W$d5-;v$K4X$9$kMWLs$,I=<($5$l!"(J
$@%a%C%;!<%89T$K$O(J{\tt vin}$@$+$i$N%a%C%;!<%8$,I=<($5$l$^$9!#(J
$@3F%l%Y%k$G$N2hLL$N9=@.$HI=<($N0UL#$r(J\ref{nandakorya2}$@$K<($7$^$9!#(J
$@%0%k!<%W%j%9%H$H5-;v%j%9%H$N%l%Y%k$G$O!"%?%$%H%k9T$H%a%C%;!<%89T$O6&DL$G$9!#(J
$@?^$K<($9$H$*$j!"A4$F$N%l%Y%k$K$*$$$F%a%C%;!<%89T$N:8B&$K!"8=:_CO!"$D$^$j8=:_(J
$@$I$N%0%k!<%W$^$?$O5-;v$NCf$K$$$k$+$,I=<($5$l$F$$$^$9!#(J

\begin{figure}[t]
\begin{center}
\makebox{\epsfile{file=/net/etlcom/usr/users/etl/guide93/vin.fig1.eps,height=18cm}}
\caption{{\tt vin}$@$NI=<(3,AX$H4pK\%3%^%s%I(J}
\label{nandakorya1}
\end{center}
\end{figure}

\begin{figure}[t]
\begin{center}
\makebox{\epsfile{file=/net/etlcom/usr/users/etl/guide93/vin.fig2.eps,height=18cm}}
\caption{{\tt vin}$@$NI=<(2hLL$N9=@.$H7A<0(J}
\label{nandakorya2}
\end{center}
\end{figure}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (4) $@%*%s%i%$%s%I%-%e%a%s%H(J}
\index{vin!$@%*%s%i%$%s%I%-%e%a%s%H(J}

{\tt vin}$@$N%^%K%e%"%k!&%I%-%e%a%s%HN`$O!"(J{\tt vin}$@$NCf$G%*%s%i%$%s$G8+$i$l$k$h$&$K(J
$@$J$C$F$$$^$9!#(J{\tt vin}$@$N2hLL$N%a%C%;!<%89T$K%G%U%)%k%H$G(J\\
\hspace*{1cm} \verb+$@!V(Jhit ^H(back-space) for help$@!W(J+\\
$@$HI=<($5$l$F$$$k$H$*$j!"(J\verb+^H+(Control{\tt +H}$@!a%P%C%/%9%Z!<%9(J)$@$r2!$9$H!"(J\\
\hspace*{1cm}
\verb+$@!V(JHELP  i:intro  s:summary  k:describe key  o:desc-object  m:manual$@!W(J+\\
$@$N$h$&$K%a%K%e!<$,=P$^$9!#(J{\tt i}$@$rBG$F$P4pK\E*$J%3%^%s%I$N0lMw(J($@>e=R$NI,?\%3%^%s%I(J
$@$N0lMw(J)$@$,!"$^$?(J{\tt s}$@$HBG$F$PA4%3%^%s%I$N0lMw$,I=<($5$l$^$9!#(J

$@$^$?!"(J{\tt cosmos.vin}\index{cosmos.vin}$@$H$$$&%0%k!<%W$N2<$K$O!"%^%K%e(J
$@%"%k(J({\tt manual})$@!"%$%s%9%H!<%k$N<j0z(J({\tt install})$@!"$=$7$F(J{\tt vin}
$@$NJQ99MzNr(J({\tt history})$@!"$H$$$&%5%V%0%k!<%W$,$"$j$^$9!#(J
$@:G?7$N5!G=$K$D$$$F$O!"(J{\tt manual}$@$K5-=R$5$l$F$*$i$:!"(J{\tt history}$@$K$7$+$J$$>l9g$,(J
$@$"$j$^$9!#=i$a$F(J{\tt vin}$@$r5/F0$7$?;~$K8+$($k>pJs$O!"(J{\tt cosmos.vin.manual}$@$NCf$N(J
$@5-;v$G$"$j!"(J\verb+^Hi+$@$O$3$N5-;v$X$N%8%c%s%W%3%^%s%I$G$9!#(J

UNIX$@$N%*%s%i%$%s%^%K%e%"%k$H$7$F$b!"K\=q$NIUO?$KE:IU$9$k$b$N$,EPO?$5$l$F$*$j!"(J
\verb+%man vin+$@$G8+$k$3$H$,$G$-$^$9!#$?$@$7!"F|K\8l$G5-=R$5$l$F$$$^$9$N$G!"(J
\verb+%setenv PAGER kless+$@$N$h$&$K$7$F$+$iFI$`$3$H$r$*4+$a$7$^$9!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (5)$@%a!<%k!&%K%e!<%9$NAw?.(J}

$@%a!<%k$d%K%e!<%9$N:n@.$HAw?.$O!"0J2<$N$h$&$J<j=g$G9T$J$$$^$9!#(J
\begin{description}
\itemsep=0mm
\item{\large $@%"(J) $@Aw?.$9$k%K%e!<%9$^$?$O%a!<%k$N5-;v$N$b$H$r:n$k(J}
\begin{verbatim}
$@!!!!!!8=:_$N5-;v$K%a!<%k$GJV?.$9$k;~!!!!(JR   (Reply)
$@!!!!!!8=:_$N5-;v$K%K%e!<%9$GJV?.$9$k;~!!(JF   (Followup)
$@!!!!!!?7$7$/%a!<%k$r:n$k;~!!!!!!!!!!(J  $@!!(Jxcm (eXtended-command$@!"(J Compose$@!"(J Mail)
$@!!!!!!?7$7$/%K%e!<%9$r:n$k;~!!!!!!!!(J  $@!!(Jxcn (eXtended-command$@!"(J Compose$@!"(J News)
\end{verbatim}
$@$3$l$i$N%3%^%s%I$K$h$j!"5-;v$,:n$i$l$F%(%G%#%?$,8F$S=P$5$l$^$9!#8F$S=P$9(J
$@%(%G%#%?$N%G%U%)%k%H$O(J{\tt nemacs -nw}$@$G$9$,!"$3$l$O4D6-JQ?t(J{\tt EDITOR}$@$G@_Dj$G$-$^$9!#(J
{\tt vi}$@%f!<%6$K$O!"F|K\8lBP1~$N(J{\tt vi}($@!V(Jonew$@$N;H$$J}(J \ref{sec:onew}$@!W;2>H(J)$@$,(J
$@$*A&$a$G$9!#(J
$@0J2<%$(J)$@!A%((J)$@$^$G!"%a!<%k$+%K%e!<%9$+$K$h$i$:<j=g$O6&DL$G$9!#(J

\item{\large $@%$(J) $@%(%G%#%?$G5-;v$rJT=8$9$k(J}\\
$@%a!<%k$N>l9g$O%X%C%@Cf$N(J{\tt To$@$d(JSubject$@!"%K%e!<%9$N>l9g$K$O(JNewsgroups$@$d(J
Subject$@$r=q$$$F(J}($@4{$K$"$k>l9g$O3NG'$7$F(J)$@2<$5$$!#K\J8$r=q$$$F!"%(%G%#%?$r(J
$@=*N;$7$F(J{\tt vin}$@$KLa$j$^$9!#(J

\item{\large $@%&(J) $@3NG'$*$h$S=$@5(J}\\
{\tt vin}$@$KLa$C$?8e!"Aw?.$9$kA0$K5-;v$rD/$a$F2<$5$$!#$b$7D>$7$?$$ItJ,$,$"$l$P(J
{\tt e}(edit$@$N0U(J)$@$rBG$F$P!"%$(J)$@$KLa$j$^$9!#(J

\item{\large $@%((J) $@Aw?.$^$?$O<h$j;_$a(J}\\
$@Aw?.$9$k;~(J:
$@BgJ8;z$G(J{\tt S}(Send$@$N0U(J)$@%3%^%s%I$rBG$D$H!"Aw?.$5$l$^$9!#Aw?.$NA0$K!"08@h$N(J
$@3NG'$r5a$a$FMh$^$9$+$i!"3NG'$7$F(J{\tt y}$@$HBG$C$F2<$5$$!#(J
$@Aw?.=*N;8e!"(J{\tt RET}$@$rBG$D$H!"$b$H$N5-;v(J($@$"$k$$$O%"(J)$@$r3+;O$7$?>l=j(J)$@$XLa$j$^$9!#(J\\
$@Cf;_$9$k;~(J:
$@Aw?.$r<h$j;_$a$k;~$K$O!"(J{\tt a}(abort$@$N0U(J)$@$rBG$D$H!"$b$H$N5-;v$KLa$j$^$9!#(J

\item{\large $@%*(J) $@<h$j2<$2$k(J}\\
$@0lC6Ej9F$7$?%K%e!<%95-;v$r:o=|$9$k$K$O!"Ej9F$5$l$?5-;v$NCf$GBgJ8;z$G(J
{\tt C}(Cancel$@$N0U(J)$@$HBG$C$F2<$5$$!#3NG'$r5a$a$F$-$^$9$N$G!"(J{\tt y}$@$HBG$C$F2<$5$$!#(J
$@$J$*!"%a!<%k$K4X$7$F$O!"0lC6Aw?.$7$?%a!<%k$r<h$j>C$9$3$H$O!"8=:_$N(J
$@%a!<%k%7%9%F%`$G$O$G$-$^$;$s!#(J
\end{description}

$@0J>e$N$h$&$K$7$F0lC6:n@.$7$?5-;v$O!"(J{\tt DRAFT}\index{vin!DRAFT}
$@$H$$$&%0%k!<%W(J($@<BBN$O!"(J\verb+~/Mail/drafts+
$@$H$$$&%G%#%l%/%H%j(J(MH$@7A<0%U%)%k%@(J))$@$N2<$KJ]B8$5$l$F$$$^$9!#;~!9$3$3$rGA$$$F!"(J
$@ITMW$J$b$N$O:o=|$7$F2<$5$$!#(J
$@Aw?.$7$?%a!<%k$O%G%U%)%k%H$G!"(J
\verb+<Mail>.record+
$@$H$$$&%0%k!<%W$K5-O?$5$l$F$$$^$9!#(J
$@$^$?!"Ej9F$7$?%K%e!<%9$r<+F0E*$KJ]B8$7$?$$>l9g$O!"(J{\tt AUTHORCOPY}$@$H$$(J
$@$&4D6-JQ?t(J\index{AUTHORCOPY}
$@$G%U%)%k%@$N%U%!%$%kL>$r@_Dj$7$F2<$5$$(J($@>\$7$/$O(J{\tt rn$@$^$?$O(Jvin}$@$N%^%K%e(J
$@%"%k;2>H(J)
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (6) $@5-;v$N0u:~(J}

$@0u:~$7$?$$5-;v$NCf$G!"(J{\tt xpp RET} $@$HF~NO$7$^$9!#:G=i$N(J{\tt xp}$@$G!"%W%j%s%H%3%^%s%I(J
$@%b!<%I$KF~$j$^$9!#$3$N%l%Y%k$G(J($@%a%C%;!<%89T$K%a%K%e!<$,I=<($5$l$k$H$*$j(J)
{\tt v}$@%3%^%s%I$K$h$C$F4v$D$+$N0u:~7A<0$rA*Br$9$k$3$H$,$G$-$^$9!#$5$i(J
$@$K(J{\tt p}$@$rBG$D$H!"(J\\
\hspace*{1cm} \verb+Printer: |/usr/ucb/lpr+\\
$@$N$h$&$KI=<($5$l$F!"(J{\tt RET}$@F~NOBT$A$K$J$j$^$9!#%G%U%)%k%H$G$J$$%W%j%s%?$K(J
$@=PNO$9$k>l9g$K$O!"(J\\
\hspace*{1cm} \verb+Printer: |/usr/ucb/lpr -P$@%W%j%s%?L>(J+\\
$@$N$h$&$K;XDj$7$F$+$i!"(J{\tt RET}$@$rBG$C$F2<$5$$!#$J$*!"4D6-JQ?t(J{\tt PRINTER}$@$r@_Dj$7$F(J
$@$*$1$P!"$=$l$,%G%U%)%k%H$H$7$F;H$o$l$^$9!#(J\index{PRINTER}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (7) $@%a!<%k$N@0M}(J($@%U%)%k%@$N4IM}(J)}

$@<u?.$7$?%a!<%k$O!"(J
\verb+<Mail>.unread+
$@$H$$$&L>A0$N%0%k!<%W$KF~$l$i$l$F$$$^$9(J
($@$3$N%0%k!<%W$N<BBN$O!"(J"{\tt /var/spool/mail/}$@$"$J$?$N%m%0%$%sL>(J"$@$H$$$&!"(J
$@%7%9%F%`$N6&M-NN0h$K$"$k%U%!%$%k(J(MBOX$@7A<0%U%)%k%@(J))$@!#(J

\begin{description}
\itemsep=0mm
\item{\large $@%"(J) $@%a!<%k$N:o=|(J}\\
$@ITMW$J%a!<%k$r:o=|$9$k>l9g$K$O!"5-;v%j%9%H$"$k$$$O5-;v%l%Y%k$G!"(J
{\tt D}(Delete)$@$H2!$7$F!":o=|0u$rIU$1$F2<$5$$!#$9$k$H!"%j%9%H%l%Y%k$G$O5-;v(J
$@$N9T$N@hF,(J(\ref{nandakorya2}$@Cf$N(JStat$@$N0LCV(J)$@$K!"(J{\tt D}$@$H$$$&%^!<%/$,IU$-$^$9!#(J
$@5-;v%l%Y%k$G$O!"3F<o%U%i%0$NCf$K!"(J{\tt D}$@$HI=<($5$l$^$9!#(J
{\tt D$@%^!<%/$r<h$j>C$9$K$O!":F$S(JD$@$HBG$C$F2<$5$$!#(J}
$@$3$N$h$&$K:o=|%^!<%/$rIU$1$?8e!"5-;v%j%9%H%l%Y%k$G!"(J{\tt r}(refresh)$@$rBG$D$H!"(J
$@<B:]$K:o=|$,9T$J$o$l!"%j%9%H$,99?7$5$l$^$9!#(J

\item{\large $@%$(J) $@%U%)%k%@$X$N@0M}J]B8(J}\\
$@%7%9%F%`$N6&M-NN0h$K%a!<%k$r;D$7$FCV$/$3$H$O?d>)$5$l$^$;$s$7!"$3$N(J
\verb+<Mail>.unread+
$@$H$$$&%U%!%$%k$,5pBg2=$9$k$H!"MxMQ<T$H$7$F$b07$$$K$/$/$J$j$^$9(J
($@FC$K!"?7Ce%a!<%k$,$"$C$?;~$K%$%s%G%C%/%9%U%!%$%k$r:n$jD>$9$N$,CY$/$J$k(J)$@!#(J
$@$3$N$?$a!"%a!<%k$NOCBj$d!"Aw?.<T$J$I$K1~$8$F!"J#?t$N!V%a!<%k%U%)%k%@(J(mail
 folder)$@!W$KJ,N`@0M}$7$FCV$/$3$H$r$*4+$a$7$^$9!#(J

{\large $@=`Hw(J}:
$@$b$H$b$H(J{\tt .mailrc$@$K(Jfolder}$@$rDj5A$7$F$$$??M$O!"0J2<$OITMW$G$9!#$^$:!"(J
\index{.mailrc}
$@%U%)%k%@$rCV$/%G%#%l%/%H%j$r:n@.$7$^$9!#Nc$($P(J{\tt mail}$@$H$$$&L>A0$K$7$^$9!#(J\\
\hspace*{1cm} \verb+% mkdir ~/mail +\\
$@$=$7$F!"(J{\tt .mailrc}$@$NCf$K!"$=$N%G%#%l%/%H%j$r0J2<$N$h$&$KEPO?$7$^$9!#(J\\
\hspace*{1cm} \verb+set folder=~/mail+\\
$@0J>e$K$h$C$F!"(J\verb+<Mail>.folders+$@$H$$$&%0%k!<%W$,:n@.$5$l!"(J
$@$=$N%5%V%0%k!<%W$H$7$F(J\verb+~/mail+$@$N2<$N%U%)%k%@72$,8+$($k$h$&$K$J$j$^$9!#(J

{\large $@J#<L(J}:
$@5-;v%j%9%H$"$k$$$O5-;v%l%Y%k$G!"(J{\tt s}(save)$@$^$?$O(J{\tt t}(transfer)$@%3%^%s%I$K$h$j(J
$@;XDj$9$k%U%)%k%@$XJ#<L$7$^$9(J({\tt t}$@$G$O!"(Jsave$@$HF1;~$K(J{\tt D}$@%^!<%/$rIU$1$k(J)$@!#(J
$@Nc$($P!"8=:_$N5-;v$r(J\verb+~/mail/sample+$@$H$$$&%U%)%k%@$K0\$9$K$O!"(J\\
\hspace*{1cm} \verb+t~/mail/sample$@!!!!!!!!!!!!!!!!!!!!(J+
$@$HF~NO$7$^$9!#>e=R$N=`Hw$,$7$F$"$l$P!"(J{\tt t}$@$HBG$C$?CJ3,$G(J\\
\hspace*{1cm} \verb+=== Folder ===>> ~/mail/$@!!!!!!!!(J+
$@$H=P$F$-$^$9$+$i!"$"$H$O(J{\tt sample}$@$HF~NO$7$F2<$5$$!#(J\\
$@$b$7(J{\tt sample}$@$,B8:_$7$J$$(J($@:#:n$m$&$H$7$F$$$k(J)$@%U%)%k%@$G$"$k$J$i!"%a%C%;!<%89T$G(J\\
\hspace*{1cm} \verb+~/mail/sample create ? (y:es)$@!!(J+
$@$N$h$&$K3NG'$r5a$a$FMh$^$9$+$i!":n$C$F$h$$>l9g$K$O(J{\tt y$@!"$=$&$G$J$$;~$O(Jy$@0J30(J}
$@$rBG$C$F2<$5$$!#(J
\end{description}

\end{description}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{vin$@$rMxMQ$G$-$k%[%9%H(J}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

$@:G$b<j4V$,$+$+$i$J$$$N$O!"6&MQ%^%7%s(J{\tt etlcom}$@$N>e$G%K%e!<%9$d%a!<%k$rFI$_=q$-(J
$@$9$k$3$H$G$9!#8=:_$N$H$3$m(J{\tt etlcom}$@$O(JCPU$@$b%G%#%9%/$bM>M5$,$"$j$^$9!#$?$@$7!"(J
$@D94|E*$K8+$FIi2YJ,;6$N$?$a$K!"8&5f<<%m!<%+%k$N%^%7%s$G(J{\tt vin}$@$r;H$&$3$H$b?d>)(J
$@$5$l$F$$$^$9!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{description}
\itemsep=0mm
\item{\large \bf (1) $@>pJs6&MQ$N%[%9%H(J}\\
{\tt vin}$@$O!"A4$F$N>pJs6&MQ%^%7%s$K!"(J{\tt /usr/local/bin/vin}$@$H$7$FEPO?$5$l$F$$$^$9!#(J
$@$3$l$i$N%[%9%H$G$O!"MxMQ<T$O2?$b8D?ME*$J@_Dj$r9T$J$&$3$H$J$/!"(J{\tt vin}$@$rMxMQ(J
$@$G$-$^$9!#$?$@$7!"D94|E*$K8+$F$*A&$a$G$-$k%[%9%H$O!"(J{\tt etlcom}$@$@$1$G$9!#(J
{\tt vin}$@$O(JCPU$@$d%a%b%j$rB?$/>CHq$9$k%D!<%k$G$9$N$G!"%K%e!<%9%5!<%P(J
($@8=:_$O(J{\tt etlss2})$@$N>e$G$O!"B>$N%^%7%s$N%K%e!<%9MxMQ<T$NLBOG$H$J$j$^$9$+$i!"(J
$@$3$3$G$O(J{\tt vin}$@$r;HMQ$7$J$$$h$&$*4j$$$7$^$9!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (2) $@>pJs7O8&5f<<%m!<%+%k%^%7%s(J}\\
$@I8=`4D6-@_Dj!J(J\ref{sec:standard-user-env}$@!K$r;HMQ$7$F$$$k>l9g$O!"8&5f<<%m!<%+%k$N(J
Sun(Sparc$@7O(J)$@$+$i!"2?$b@_Dj$J$7$K(J{\tt vin}$@$H$9$k$@$1$G;H$($^$9!#$=$l0J(J
$@30$N>l9g$K$O!"0J2<$N@_Dj$r$7$^$9!#(J
$@%m!<%+%k%^%7%s$HF1$8%"!<%/%F%/%A%c$H(JOS$@$r;}$D6&MQ%^%7%s$r(J{\tt SERVER}$@$H$7$^$9!#(J
$@6&MQ(J{\tt SERVER}$@>e$N(J{\tt vin}$@$r%^%&%s%H$"$k$$$O%3%T!<$7$F;H$&$3$H$K$J$j$^$9!#(J
$@8=:_>pJs6&MQ$G$O(JSparc$@7O$N$_$rDs6!$7$F$*$j!"(J{\tt SERVER=etlfs}$@$H$J$C$F$$$^$9!#(J
(SUN3$@!"(JDECstation$@!"(JSONY-NEWS$@MQ$N(J{\tt vin}$@$b$"$j$^$9!#;H$$$?$$J}$O(J
$@:n@.<T(J({\tt ysato@etl.go.jp})$@$^$G%a!<%k$GLd$$9g$o$;$F2<$5$$(J)
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{description}
\itemsep=0mm
\item{\large $@%"(J) $@%^%&%s%H$7$F;H$&>l9g(J}\\
{\tt automount$@$r@_Dj$7$F2<$5$$(J}
($@K\%,%$%I!V(J\ref{sec:install}
 $@?7$7$$%o!<%/%9%F!<%7%g%s$N@_Dj$H0];}!W;2>H(J)$@!#(J
$@$9$k$H!"C1$K(J{\tt /net/etlfs/usr/local/bin/vin}$@$H$7$F;HMQ$9$k$3$H$,$G$-$^$9!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large $@%$(J) $@%3%T!<$7$F;H$&>l9g(J}\\
LAN$@$N4X78$G(JNFS$@$,CY$$>l9g$K$O!"%m!<%+%k$K%3%T!<$7$F;H$C$F$b$+$^$$$^$;$s!#(J
$@4pK\E*$K$O(J
\dense
\begin{verbatim}
$@!!(J  local# mkdir /usr/local/cosmos; chown $USER /usr/local/cosmos
$@!!(J  local% rsh -n SERVER "cd /usr/local/cosmos; tar cf - cosmos .COSMOS" \
$@!!(J                      |(cd /usr/local/cosmos; tar xf -)
$@!!(J  local# ln -s /usr/local/cosmos/cosmos /usr/local/bin/vin
\end{verbatim}
\sparse
$@$G;H$($^$9!#I,MW$J%G%#%9%/NL$O!"Ls(J4M$@%P%$%H$G$9!#(J
$@$?$@$7!"%3%T!<$7$?HG$,?7A/$J%P%0F~$j$G$"$k$3$H$b$"$j$^$9$+$i!"Cm0U$7$F(J
$@2<$5$$!#$^$?!"%3%T!<$7$?>l9g$K$O$=$N;]!":n@.<T$^$G%a!<%k$GCN$i$;$F2<$5$$!#(J
\end{description}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large \bf (3) $@>pJs6&MQ(JLAN$@0J30$K@\B3$7$F$$$k%[%9%H(J}\\
$@$*$=$i$/(J(2-$@%$(J)$@$N%3%T!<$r9T$J$&I,MW$,$"$k$G$7$g$&!#$^$?!"%M%C%H%o!<%/7PM3(J
$@$G%K%e!<%9$rFI$`$?$a$N(JNNTP$@%5!<%S%9$,$=$N%[%9%H$KBP$7$FDs6!$5$l$F$$$J$$(J
$@2DG=@-$,$"$j$^$9!#(J
$@$=$N>l9g$K$O!"%]%9%H%^%9%?(J({\tt postmaster@etl.go.jp})$@$^$GLd$$9g$o$;$F2<$5$$!#(J
\end{description}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{$@J#?t$N%[%9%H4V$N%a!<%k!&%K%e!<%94D6-6&M-$NJ}K!(J}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

$@J#?t$N%^%7%s$rEO$jJb$$$F;E;v$r$9$k%f!<%6$O!"$I$N%^%7%s$+$i$bF1$84D6-$G!"(J
$@%K%e!<%9$d%a!<%k$rFI$a$k$h$&$K$7$FCV$/$HJXMx$G$9!#$=$N$?$a$K$O!"0J2<$N(J
$@@_Dj$r9T$J$C$F2<$5$$!#(J
\dense
\begin{verbatim}
$@!!!&%a!<%k$d4IM}%U%!%$%k$rCV$/%[%9%H$r7h$a$k(J($@0J2<(JHOMEHOST$@$H$9$k(J)
$@!!!&(JDOTDIR$@4D6-JQ?t$G!"4IM}%U%!%$%k$NCV$->l=j$r(J(HOMEHOST$@>e$K(J)$@;XDj$9$k!#(J
$@!!!&$I$N%^%7%s$+$i$b(JHOMEHOST$@$K(Jrsh$@$G$-$k$h$&!"(J.rhosts$@$r=q$/!#(J
$@!!!&(JHOMEHOST$@>e$N(J{\tt vin}$@$N4D6-@_Dj%U%!%$%k(J(.vin_setup)$@$r=q$/!#(J
$@!!!&%a!<%k%U%)%k%@$r(JHOMEHOST$@>e$K=8Cf$9$k(J(.mailrc + symbolic-link)
\end{verbatim}
\sparse
$@$J$*!"0J2<$N@_Dj$K$h$j!"(J{\tt rn}$@$*$h$S(JUCB mail$@$N4D6-$b!"(J
{\tt HOMEHOST}$@$KE}9g$5$l$^$9!#(J
\begin{description}
\itemsep=0mm
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large (0) {\tt HOMEHOST}$@$NA*Br(J}\\
$@6&MQ7W;;5!$r;H$&>l9g!"(J{\tt HOMEHOST$@$H$7$F$O!"(Jetlcom$@$r?d>)$7$^$9!#(J}
$@8&5f<<%m!<%+%k$N%^%7%s$r(J($@6&MQ%^%7%s$+$i8+$F(J){\tt HOMEHOST}$@$K$7$J$$$G2<$5$$!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large (1) {\tt .cshrc}\index{.cshrc}$@$K(J{\tt DOTDIR}$@$rDj5A(J}\\
{\tt vin}$@$rMxMQ$9$k3F%^%7%s$N(J{\tt .cshrc}$@$K0J2<$N$h$&$K@_Dj$7$^$9!#(J
\begin{verbatim}
$@!!!!(Jsetenv DOTDIR /net/HOMEHOST/MYHOME
\end{verbatim}
$@$3$3$G(J{\tt MYHOME}$@$O!"B>$N%^%7%s$+$i(JNFS$@$G%"%/%;%9$7$^$9$+$i!"%7%s%\%j%C%/(J
$@%j%s%/$r4^$^$J$$L>A0$K$7$F2<$5$$!#Nc$($P!"(J\verb+MYHOME=/usr/usrs/$USER+$@$O(J
$@%@%a!"(J\verb+/home/ss22/$USER+$@$J$i#O#K!"$G$9!#(J
$@$J$*!"3F%^%7%s$G4{$K%[!<%`%G%#%l%/%H%j$r6&DL2=$7$F$$$k>l9g(J
$@$K$O!"$3$N@_Dj$OITMW$G$9!#(J
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large (2) {\tt .rhosts}}\\
{\tt $@3F%^%7%s$N(J.rhosts$@$K(JHOMEHOST$@$r2C$($k!#(J}\\
{\tt HOMEHOST$@$N(J.rhosts$@$K3F%^%7%s$r2C$($k!#(J}\index{.rhosts}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large (3) {\tt .vin\_setup}}\\
{\tt HOMEHOST}$@$N!"(J\verb+$DOTDIR/.vin_setup+$@$K!"0J2<$N$h$&$K=q$-$^$9!#(J
\dense
\begin{verbatim}
$@!!!!(J//////// $@%a!<%k!&%K%e!<%94D6-$r(JHOMEHOST$@>e$K=8Cf$9$k(J //////////////////
$@!!!!(JHOMEHOST: '$@$3$3$K(JHOMEHOST$@$NL>A0$r=q$$$F$M(J'$@!!!!!!!!!!!!(J## $@Nc(J: etlcom
$@!!!!(JMAIL_SPOOL:  '/net/'<HOMEHOST> '/var/spool/mail'
$@!!!!(JMAIL_POSTER: 'rsh  '<HOMEHOST> ' /usr/lib/sendmail -t -om'
$@!!!!(J//////////////////////////////////////////////////////////////////////
\end{verbatim}
\sparse
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\item{\large (4) $@%U%)%k%@$NE}9g(J}\\
$@6&DL$N%U%)%k%@$r!"(J{\tt HOMEHOST}$@$K:n$j$^$9!#Nc$($P!"(J\verb+~/mail+$@$H$7$^$9!#$=$7$F$=$N(J
$@%G%#%l%/%H%j$N2<$K3F<o%U%)%k%@$r=8Cf$9$k$h$&(J{\tt .mailrc}$@$KDj5A$7$^$9!#(J
\index{.mailrc}
\dense
\begin{verbatim}
$@!!!!(JHOMEHOST% mkdir ~/mail
$@!!!!(JHOMEHOST% $DOTDIR/.mailrc$@Cf$N(Jfolder$@4X78$NDj5A$r!"0J2<$N$h$&$KJQ$($k!#(J
$@!!!!!!(Jset folder=~/mail
$@!!!!!!(Jset MBOX=~/mail/mbox
$@!!!!!!(Jset record=~/mail/record
\end{verbatim}
$@<!$K!"3F%^%7%s$N(J{\tt mail$@$H(J.mailrc$@$r!"(JHOMEHOST$@$K%j%s%/$7$^$9!#(J}
\begin{verbatim}
$@!!!!(Jlocal% ln -s $DOTDIR/mail $HOME
$@!!!!(Jlocal% ln -s $DOTDIR/.mailrc $HOME
\end{verbatim}
\sparse
{\tt $@>e=R$N(JDRAFT$@%U%!%$%k$J$I$b!"(JHOMEHOST$@$KE}9g$9$k>l9g$K$O!"0J2<$N$h$&$K(J
$@%j%s%/$7$^$9!#(J}
\begin{verbatim}
$@!!!!(Jlocal% ln -s $DOTDIR/Mail $HOME
\end{verbatim}
\end{description}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\index{DOTDIR}
$@>e=R$N(J{\tt DOTDIR$@$O!"(Jvin$@$d(Jrn}$@$N4IM}%U%!%$%k(J({\tt .vinrc$@$d(J.newsrc}$@$J$I(J
$@$N(Jdot-file)$@$rJ]B8$9$k%G%#%l%/%H%j$r;XDj$7$F$$$^$9!#$3$l$O!"F10l%^%7%s(J
$@$GJ#?t$N(J{\tt vin}$@4D6-$r;H$$J,$1$k$?$a$K$b;H$&$3$H$,$G$-$^$9!#(J
\nextguide